Author's response to reviews

Title: Implementation of geriatric assessment and decision support as part of a Multidisciplinary Integrated Care model in residential care homes

Authors:

Marijke Boorsma Md (m.boorsma@vumc.nl)
eveliene langedijk ms (eveliene_langedijk@hotmail.com)
dinnus h.m frijters phd (d.frijters@vumc.nl)
giel njipels prof (g.nijples@vumc.nl)
tom elfring prof (t.elfring@vu.nl)
hein p.j hout phd (hpj.vanhout@vumc.nl)

Version: 2 Date: 27 July 2011

Author's response to reviews: see over
Dear Dr Appleford,

Thank you for your message. We cited the study protocol on page 3 and the results of the trial are in press by CMAJ. We added the following sentences on page 6 on behalf of the ethics-approval and consent: The study was approved by the ethics committee of the VU University Medical Center in Amsterdam, the Netherlands. All participating residents or their proxies provided written informed consent.

On behalf of the co-authors,

Yours Sincerely,

Marijke Boorsma